CSC 405, Week 3 Requirements and Acceptance Test Discussions, (Postponed from Week 2) Overview of Testing Frameworks

I. Weekly Lecture/Lab Overview A. Monday: 1. Individual requirements reports 2. Individual acceptance testing reports 3. Details of phase 1 work assignment

B. Wednesday: 1. Intro to functional testing frameworks 2. Lab time for development and testing

C. Friday: 1. More on functional testing frameworks 2. Lab time for development and testing

II. Discussion of Requirements Updates A. Postponed from week 2 B. Everyone summarizes their rqmts updates C. Note any issues requiring group discussion D. Finalize phase 1 features list

III. Discussion of Initial Acceptance Testing A. Postponed from week 2 B. Everyone summarizes their testing C. Note any issues requiring group discussion D. Finalize phase 1 must-fix list

IV. Phase 1 work assignments A. See Jira postings from E & K B. Fill in wiki table C. Lab time to discuss specific work